# UC Berkeley UC Berkeley Previously Published Works

# Title

Health care services use, stillbirth, and neonatal and infant survival following implementation of the Maternal Health Voucher Scheme in Bangladesh: A difference-indifferences analysis of Bangladesh Demographic and Health Survey data, 2000 to 2016

# Permalink

https://escholarship.org/uc/item/4s8178m7

**Journal** PLOS Medicine, 19(8)

# ISSN

1549-1277

## **Authors**

Nandi, Arijit Charters, Thomas J Quamruzzaman, Amm <u>et al.</u>

# **Publication Date**

2022

# DOI

10.1371/journal.pmed.1004022

# **Copyright Information**

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed



# G OPEN ACCESS

**Citation:** Nandi A, Charters TJ, Quamruzzaman A, Strumpf EC, Kaufman JS, Heymann J, et al. (2022) Health care services use, stillbirth, and neonatal and infant survival following implementation of the Maternal Health Voucher Scheme in Bangladesh: A difference-in-differences analysis of Bangladesh Demographic and Health Survey data, 2000 to 2016. PLoS Med 19(8): e1004022. https://doi.org/ 10.1371/journal.pmed.1004022

**Academic Editor:** Zulfiqar A. Bhutta, The Hospital for Sick Children, CANADA

Received: June 2, 2021

Accepted: May 16, 2022

Published: August 15, 2022

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004022

**Copyright:** © 2022 Nandi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Health care services use, stillbirth, and neonatal and infant survival following implementation of the Maternal Health Voucher Scheme in Bangladesh: A differencein-differences analysis of Bangladesh Demographic and Health Survey data, 2000 to 2016

# Arijit Nandi<sup>1,2</sup>\*, Thomas J. Charters<sup>1</sup>, Amm Quamruzzaman<sup>3</sup>, Erin C. Strumpf<sup>1,2</sup>, Jay S. Kaufman<sup>1,2</sup>, Jody Heymann<sup>4</sup>, Arnab Mukherji<sup>5</sup>, Sam Harper<sup>1,2</sup>

1 Institute of Health and Social Policy, McGill University, Montréal, Québec, Canada, 2 Department of Epidemiology, Biostatistics and Occupational Health, Montréal, Québec, Canada, 3 Interdisciplinary Studies Field, University of California, Berkeley, California, United States of America, 4 WORLD Policy Analysis Center, Fielding School of Public Health, University of California, Los Angeles, California, United States of America, 5 Indian Institute of Management Bangalore; Centre for Public Policy, Bengaluru, India

\* arijit.nandi@mcgill.ca

# Abstract

## Background

Starting in 2006 to 2007, the Government of Bangladesh implemented the Maternal Health Voucher Scheme (MHVS). This program provides pregnant women with vouchers that can be exchanged for health services from eligible public and private sector providers. In this study, we examined whether access to the MHVS was associated with maternal health services utilization, stillbirth, and neonatal and infant mortality.

## Methods and findings

We used information on pregnancies and live births between 2000 to 2016 reported by women 15 to 49 years of age surveyed as part of the Bangladesh Demographic and Health Surveys. Our analytic sample included 23,275 pregnancies lasting at least 7 months for analyses of stillbirth and between 15,125 and 21,668 live births for analyses of health services use, neonatal, and infant mortality. With respect to live births occurring prior to the introduction of the MHVS, 31.3%, 14.1%, and 18.0% of women, respectively, reported receiving at least 3 antenatal care visits, delivering in a health institution, and having a skilled birth attendant at delivery. Rates of neonatal and infant mortality during this period were 40 and 63 per 1,000 live births, respectively, and there were 32 stillbirths per 1,000 pregnancies lasting at least 7 months. We applied a difference-in-differences design to estimate the effect of providing subdistrict-level access to the MHVS program, with inverse probability of

Data Availability Statement: Analyses utilized two sources of secondary data, specifically: (1) pregnancy and live birth information collected from respondents surveyed as part of the Bangladesh Demographic and Health Surveys (DHS) and (2) upazila-level data on socio-demographic and other indicators from the World Bank's Bangladesh Interactive Poverty Maps 2010. The DHS data are publicly available, but users must first register with the DHS program. Registration, which requires a summary of the proposed study and selection of country datasets, can be completed at: http://www. dhsprogram.com/data/new-user-registration.cfm. The World Bank data are available for download here: http://www.worldbank.org/en/news/feature/ 2014/09/30/poverty-maps The statistical code for producing the analytic dataset and replicating our results are available with unrestricted access from the Harvard Dataverse: https://dataverse.harvard. edu/dataverse/3po.

**Funding:** All authors acknowledge funding from the Canadian Institutes of Health Research Operating Grant, "Examining the impact of social policies on health equity" (ROH-115209). AN was supported by the Canada Research Chairs program. The funders had no role in study design, data collection and analysis, interpretation of data, preparation of the manuscript, or decision to submit the paper for publication.

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: CI, confidence interval; BDHS, Bangladesh Demographic and Health Survey; DD, difference-in-difference; DHS, Demographic and Health Survey; EA, enumeration area; GDP, gross domestic product; ITT, intention-to-treat; LMIC, low- and middle-income country; MHVS, Maternal Health Voucher Scheme; MOHFW, Ministry of Health and Family Welfare; SDG, Sustainable Development Goal. treatment weights to address selection into the program. The introduction of the MHVS program was associated with a lagged improvement in the probability of delivering in a health facility, one of the primary targets of the program, although associations with other health services were less evident. After 6 years of access to the MHVS, the probabilities of reporting at least 3 antenatal care visits, delivering in a health facility, and having a skilled birth attendant present increased by 3.0 [95% confidence interval (95% CI) = -4.8, 10.7], 6.5 (95% CI = -0.6, 13.6), and 5.8 (95% CI = -1.8, 13.3) percentage points, respectively. We did not observe evidence consistent with the program improving health outcomes, with probabilities of stillbirth, neonatal mortality, and infant mortality decreasing by 0.7 (95% CI = -1.3, 2.6), 0.8 (95% CI = -1.7, 3.4), and 1.3 (95% CI = -2.5, 5.1) percentage points, respectively, after 6 years of access to the MHVS. The sample size was insufficient to detect smaller associations with adequate precision. Additionally, we cannot rule out the possibility of measurement error, although it was likely nondifferential by treatment group, or unmeasured confounding by concomitant interventions that were implemented differentially in treated and control areas.

#### Conclusions

In this study, we found that the introduction of the MHVS was positively associated with the probability of delivering in a health facility, but despite a longer period of follow-up than most extant evaluations, we did not observe attendant reductions in stillbirth, neonatal mortality, or infant mortality. Further work and engagement with stakeholders is needed to assess if the MHVS has affected the quality of care and health inequalities and whether the design and eligibility of the program should be modified to improve maternal and neonatal health outcomes.

### Author summary

#### Why was this study done?

- Most women in Bangladesh do not receive essential maternal health services, with especially low rates of coverage for poorer households in rural areas.
- Starting in 2006 to 2007, the Government of Bangladesh implemented the Maternal Health Voucher Scheme (MHVS) to increase demand for and equitable use of maternal health services.
- Prior studies suggest that access to the MHVS, consistent with the broader literature on maternal voucher programs, was associated with greater use of priority maternal health services.
- However, it is unclear if short-term increases in maternal health service use are sustained over time and evidence for impacts on perinatal, infant, and maternal health outcomes, including mortality, is inconclusive.

#### What did the researchers do and find?

- We linked details on the rollout of the MHVS across subdistricts (i.e., upazilas) to information on pregnancies and live births reported by women surveyed as part of the Bangladesh Demographic and Health Surveys between 2000 to 2016.
- We used a difference-in-differences design to evaluate the association between upazilalevel access to the MHVS and maternal health services utilization, stillbirth, and neonatal and infant mortality.
- We observed increases in the use of maternal health services, particularly on the probability of delivering in a health facility, that materialized 2 or more years after program implementation.
- However, improvements in stillbirths, neonatal, and infant mortality were not demonstrated, raising important questions about program implementation.

#### What do these findings mean?

- Potential explanations of the gap between increased services and improved outcomes include that the program may not have been targeted to reach the highest risk mothers, that implementation may have failed to increase use by mothers at highest risk, that in the absence of simultaneous supply-side interventions quality of services provided may have declined with increased numbers of patients or been inadequate, and that hospital births are associated with higher rates of bottle feeding in Bangladesh and lower rates of protective breastfeeding.
- Additionally, it is possible that small but consequential effects in health outcomes could not be discerned due to imprecision in these estimates.
- Further research is needed to examine why increased services have not yet translated into improved health outcomes as it pertains to the MHVS and other voucher programs.
- This could include cluster randomized experimental designs with the potential to address the main limitations of this study, specifically the potential for observational studies to conflate the impact of a voucher program with the effects of other population-level health and social interventions.

#### Introduction

Progress in population health in Bangladesh has been called a paradox. Despite high rates of poverty, low per capita gross domestic product (GDP), and modest health systems investments [1,2], the country has made substantial progress in health since its independence in 1971, including dramatic reductions in under 5 and maternal mortality [1,3]. Despite its impressive progress, rates of maternal and neonatal mortality, estimated at 173 per 100,000 live births in 2017 and 17 per 1,000 live births in 2018 [4,5], respectively, remain high in Bangladesh compared to 2030 Sustainable Development Goal (SDG) targets. Similarly, many women are not

receiving essential maternal health services, with especially low rates of coverage for poorer households in rural areas [6–8]. According to the 2017 to 2018 Demographic and Health Surveys (DHS), among women ages 15 to 49 years who reported a live birth in the preceding 3 years, only 44% received 4 or more antenatal care visits, 49% delivered in a health facility, and 53% were assisted at delivery by medically trained personnel [9]. Further improvements in maternal and newborn health necessary to achieve the SDGs in these areas will require expanded coverage of quality reproductive and maternal health interventions that reach underserved populations [10,11].

Supply-side funding has been the primary strategy for increasing access to health services in Bangladesh, with the government providing funds to public facilities to provide free or highly subsidized maternal care [12]. Nonetheless, access is not universal, due in part to the inadequate supply of skilled health professionals [13], unofficial user fees, and discrimination by public providers against poor women [14,15], with many turning to private facilities [16]. Further supply-side investment is unlikely to reach socioeconomically disadvantaged groups without accompanying efforts that address demand-side barriers [2,8,11]. These barriers include distance to facilities, as well as travel costs and other out-of-pocket expenditures (e.g., medicines) that are not covered [16,17], and can be substantial even at public facilities [18,19]. Cultural beliefs and norms, lack of awareness about sources of care, and social stigma also shape demand for maternal health services [12].

Demand-side funding schemes are designed to increase the quantity and quality of maternal health services used by underserved populations by reducing financial barriers to access and increasing competition in the market. One example is Bangladesh's Maternal Health Voucher Scheme (MHVS), which provides pregnant women with vouchers that can be exchanged for health services from eligible public and private sector providers who are later reimbursed.

The MHVS was implemented by the Bangladesh Ministry of Health and Family Welfare (MOHFW), with technical and financial support from several multilateral and bilateral organizations, and aims "to increase demand and utilization of maternal health services, to improve access to and utilization of safe delivery, to encourage institutional delivery, and to improve equity in the utilization of maternal health services," with the ultimate goal of improving maternal and neonatal health outcomes [20,21]. It was introduced in 2006 to 2007 as a pilot program in 21 upazilas, which are subdistrict administrative units with a local governance structure that direct development activities within their boundaries [22]. By 2010, the MHVS expanded to cover 53 of Bangladesh's 545 upazilas (based on 2011 census boundaries), with another 2 upazilas gaining access in 2017 to 2018. According to the program's design, the MHVS was prioritized to upazilas with greater need based on their sociodemographic profile (e.g., literacy rate, poverty level), as well as the presence of health workers to help administer the program. However, in practice, the program was not geographically targeted based on poverty rates [21]. Within upazilas selected to participate in the program, individual-level access was initially universal in some districts and means-tested in others, but gradually shifted to a means-tested program in all upazilas. To enroll beneficiaries, community health workers identified pregnant women in their first trimester and applied to a local committee to determine if eligibility criteria were met based on residence in the upazila, parity, income, and land and asset ownership (Table 1). These criteria, however, were not consistently enforced. For example, vouchers were sometimes given to women who were not pregnant for the first or second time and poverty criteria were difficult to verify, resulting in imperfect targeting to poorer households [21,23]. Community health workers assisted with the distribution of program information, vouchers, referrals, subsidies, and cash incentives.

Eligibility	Benefits				
• Residence in program (treated) upazila	Three antenatal care checkups				
• Pregnancy is the first or second, with a family planning method used between pregnancies	• Delivery by skilled attendant (home or facility); beneficiaries received a Tk. 500 (USD 7.20) <sup>2</sup> incentive for delivery in home and Tk. 2,000 (USD 28.80) incentive for delivery in a facility				
• Household income Tk. 3,100 (USD 44.64) or less per month	• Management of complications including cesarean section by designated providers				
• Functionally landless (<6,534 ft <sup>2</sup> or approximately 0.15 acres of land)	• Postnatal care checkup within 6 weeks				
• Household does not own income-earning assets (e.g., livestock, rickshaw)	• Transportation costs up to Tk. 500 (USD 7.20)				
	• Gift box worth Tk. 500 (USD 7.20)				

#### Table 1. Bangladesh MHVS eligibility criteria and benefits<sup>1</sup>.

<sup>1</sup>Information from Khan and Khan (2016) [21].

<sup>2</sup>Based on a 2010 annual average exchange rate of 0.0144 USD for each Tk.

MHVS, Maternal Health Voucher Scheme; Tk., Bangladesh Taka; USD, US dollar.

https://doi.org/10.1371/journal.pmed.1004022.t001

Specific services and fees covered by vouchers included: consultation fees for up to 3 antenatal care checkups, delivery by a skilled birth attendant at the home or a health facility, management of complications including eclampsia management and delivery by cesarean section, and a single postnatal care checkup within 6 weeks of delivery. Services were covered if received from accredited public, nongovernmental, or private sector providers with the capacity to provide emergency obstetric care. Providers were reimbursed through locally assigned banks after submitting evidence of service provision. Beneficiaries were reimbursed up to 500 Taka (Tk.) (7.20 US dollars (USD) based on a 2010 annual average exchange rate of 0.0144 USD per Tk.) for transport to antenatal care visits, delivery, and postnatal care services and to cover referral fees from subdistrict to district-level facilities. Additionally, beneficiaries received a cash payment of Tk. 2,000 (USD 28.80) for delivering in a health facility or Tk. 500 (USD 7.20) for delivering at home with a skilled attendant, as well as a baby-care gift box valued Tk. 500 (USD 7.20). Although it is primarily a demand-side funding scheme, the program also aims to address the supply-side through payments to public sector facilities to increase the availability and quality of maternal health services. Further details regarding the design and operation of the MHVS are available elsewhere [20,21].

Subnational, cross-sectional comparisons of voucher recipients versus nonrecipients, as well as mothers who had recently given birth in upazilas where the program was offered versus not, suggest that the MHVS was associated with greater use of maternal health services and more comprehensive maternal care, with inconsistent evidence on out-of-pocket expenditures [18,20,24–27]. However, this research cannot inform inference regarding the causal effect of the MHVS. The only quasi-experimental evaluation, a differences-in-differences analysis that compared changes in outcomes for 11 upazilas before and after they were added to the program in 2010 to corresponding changes from 11 matched control upazilas, suggests that the program increased use of public facilities, but did not influence the proportion of women receiving maternal health services unless the treated group was restricted to 5 high-performing upazilas [28,29].

In this study, we used national information on pregnancies and live births reported by women surveyed as part of the 2004, 2007, 2011, 2014, and 2017 to 2018 Bangladesh Demographic and Health Surveys (BDHS) to evaluate the association between upazila-level access to the MHVS and maternal health services utilization, stillbirth, and neonatal and infant mortality. Our evaluation of the MHVS was designed to address several knowledge gaps concerning the impact of maternal voucher programs, including: (1) if short-term increases in the use of priority maternal health services are sustained over time [30,31]; and (2) whether they affect maternal and neonatal health outcomes [30-32], where population-level evidence is limited [30,33].

#### Methods

#### Study design

We used a difference-in-differences (DD) approach [34-36] to estimate the "intention-totreat" (ITT) effect of providing upazila-level access to the MHVS (the treatment) on our outcomes of interest. The standard DD estimate compares outcome trends before versus after an intervention with corresponding trends among a "control" group, which represents the counterfactual. Since access to the MHVS was expanded to treated upazilas over the course of approximately 4 years (i.e., variation in treatment timing), we used event study models to: (1) assess whether outcome trends in treated and control areas were parallel in the preintervention period; and (2) estimate the effect of providing access to the MHVS in treated upazilas (i.e., the average treatment effect on the treated) [37,38]. A prospective analysis plan, available in S1 Protocol, was used in designing the study, but was not preregistered or published. In contrast to the analysis plan, we added stillbirth as an outcome and did not evaluate whether the program has affected socioeconomic inequalities in our primary outcomes, due to inadequate sample sizes and the imprecision of stratified estimates. In addition, we incorporated event study models and inverse probability weights in our DD analyses to assess evidence for parallel preintervention trends and construct more comparable treatment groups. This study is reported as per the Reporting of studies Conducted using Observational Routinely-collected Data (RECORD) guidelines (S1 Checklist).

#### Data

Individual-level data were derived from the 2004, 2007, 2011, 2014, and 2017 to 2018 rounds of the BDHS, which are conducted in coordination with the worldwide DHS program [39]. The BDHS are repeated cross-sectional household surveys that provide information on the sociodemographic, health, and nutritional profile of the population [40]. The BDHS uses a stratified, multistage cluster sampling scheme based on the Bangladesh census and covers the entire population residing in noninstitutional dwellings. In the first stage, enumeration areas (EAs), typically city blocks in urban areas and villages in rural areas, are selected with the probability of selection proportional to EA size. In the second stage, 30 households were randomly sampled from selected EAs. Based on the household roster, all ever-married women (ages 15 to 49 in 2014 and 2017 to 2018, 12 to 49 in 2011, 10 to 49 in 2007, and 10 to 49 in 2004) who were usual members of a selected household were asked about their contraceptive use, reproductive history, and maternal health services use, among other factors. Comparability of data across waves is enhanced through interviewer training and standardized survey methods. Response rates for eligible women ranged between 97.9% and 98.8% across the 5 waves. Further details regarding the sampling, survey, and quality control procedures are available elsewhere [9,40-43]. For our analyses, we used information reported by women ages 15 to 49 to create samples of pregnancies and live births during the study period from 2000 to 2016. Details regarding sample selection are reported in S1 Table. The Institutional Review Board of McGill University reviewed and approved this study (IRB Study Number A05-E42-17B).

#### Measures

**Treatment.** The treatment, measured at the upazila level, was gaining access to the MHVS. The list of the 55 treated upazilas that introduced the MHVS during the study period was available through public reports [20,21,29]. However, the month and year of implementation was not recorded for all treated upazilas. Through personal communication (AQ) with government officials with direct knowledge of the MHVS, we obtained the years of implementation for each upazila. We were advised to assume a starting month of December, which is the midpoint of the financial year, since it provides time for funds allocated to the program to be dispersed and the program to be implemented. Because access to the program was not randomized across upazilas, we used upazila-level administrative data (S2 Table) from the World Bank's Bangladesh Interactive Poverty Maps [44] to model the probability of treatment and create more exchangeable control groups.

We used ArcGIS software to map the location of each respondent's EA, as a proxy for the location of their residence, to its corresponding upazila based on Global Positioning System coordinates from the center of the EA, which are provided by the BDHS. This allowed us to identify if each respondent lived in a treated or control upazila and, if treated, the timing of recorded pregnancies and live births in relation to program implementation. Fig 1 shows MHVS implementation across upazilas. We added a buffer of 2.5 km around the geographic boundaries of treated upazilas, in order to account for the random displacement of EAs in publicly available DHS data and potential spillovers, as discussed in **Appendix A** in **S1 Appendix**.



Fig 1. Implementation of the Bangladesh MHVS by year across subdistricts (upazilas) and locations of BDHS EAs, represented by points. Base layer extracted from the GADM database (www.gadm.org) version 3.6 (April 2018), available here: https://geodata.ucdavis.edu/gadm/gadm3.6/shp/gadm36\_BGD\_shp.zip. BDHS, Bangladesh Demographic and Health Survey; EA, enumeration area; MHVS, Maternal Health Voucher Scheme.

https://doi.org/10.1371/journal.pmed.1004022.g001

**Outcomes.** Indicators of maternal health services utilization, with respect to live births, included whether women reported: at least 3 antenatal care visits, which is consistent with the number of visits covered by the MHVS; having a skilled birth attendant at delivery, including a qualified doctor, nurse, midwife, paramedic, community skilled birth attendant, or family welfare visitor [45,46]; delivering in an institution or health facility outside the home, including public, nongovernmental, and private medical hospitals, clinics, and health centers; and whether the delivery was by cesarean section. We defined stillbirth as a pregnancy lasting at least 7 months that did not result in a live birth [47]. We created binary indicators for neonatal and infant mortality to record deaths occurring within the first 28 days and 1 year of life, respectively, among live births.

**Other covariates.** We included household and individual-level characteristics that might be associated with the outcomes of interest. Demographic characteristics included women's age at marriage, household size (categorized as 1 to 4, 5 to 6, or 7 or more people), rural/urban residence, and division of residence. Socioeconomic characteristics included women's and husbands' educational attainment (categorized as none, primary, or secondary or higher) and a measure of household wealth (dichotomized as above or below the median value), which is based on ownership of specific assets, environmental conditions, and housing characteristics and was constructed by DHS using the method proposed by Filmer and Pritchett [48,49]. Pregnancy and birth-related characteristics included women's age at the time of the stillbirth or live birth, measured continuously, a binary indicator for whether the interval between the index birth or pregnancy outcome and a prior birth was short (<24 months) or not (24 + months and first births), and the number of prior stillbirths.

#### Statistical analyses

Event study models are an extension of the standard 2-way fixed effects DD model in which the treatment effect is estimated by a series of leads and lags representing the timing of each observation (recorded pregnancies and live births) in relation to when the MHVS was introduced in the respondent's upazila of residence:

$$Y_{ict} = \alpha_0 + \sum_{l=\leq -6}^{-2} \mu_l A_{ct}^l + \sum_{l=0}^{\geq 6} \mu_l A_{ct}^l + \tau_p + \omega_t + \sum_{d=1}^{6} \gamma_d D_d + \beta'_j X_{ict} + \varepsilon_{ict}.$$
 (1)

In the linear probability model above, we estimated for observation *i* in upazila *c* in year *t* the effect of upazila-level access to the MHVS on the probability of each outcome,  $Y_{ict}$ , where the treatment,  $A_{ct}^l$ , is modeled as a series of 2-year event periods; the indicators for these periods, *l*, included 4 leads (i.e., 6+, 4 to 6, 2 to 4, and 0 to 2 years prior to the introduction of the MHVS) and 4 lags (i.e., 0 to 2, 2 to 4, 4 to 6, and 6+ years after the introduction of the MHVS), with the 2-year period prior to implementation taken as the reference. Observations from upazilas that did not receive access to the MHVS during the study period were assigned zeros for each indicator.

To account for unmeasured time-fixed differences between treatment groups, we included a fixed effect,  $\tau_p$ , that indicated whether the observation came from an upazila that gained access to the MHVS during the study period or not. Because the BDHS resamples upazilas in each round, we could not include upazila-level fixed effects. However, as described in **Appendix B** in **S1 Appendix**, we estimated the probability that an upazila gained access to the MHVS program during the study period, conditional on upazila-level sociodemographic characteristics, restricted the sample to the region of common support, and weighted models by the stabilized inverse probability of treatment weight [50,51]. We included fixed effects for 6 divisions (based on the 6 divisions represented in the 2004 and 2007 surveys),  $\gamma_d$ , to account for interdivision differences that might be correlated with the treatment and outcomes of interest. We also included fixed effects for year of birth (or the pregnancy outcome in analyses of stillbirth),  $\omega_t$ , to account for secular trends in outcomes during the study period shared between treated and control upazilas. We controlled for a vector of time-varying individual-level covariates,  $X_{ict}$ , taken at the time of survey and assigned to each observation in year *t*, including socioeconomic, pregnancy, and birth-related characteristics. Conditional on the other covariates in the model, the coefficient for each event period,  $\mu_b$  represents the difference in the probability of the outcome between treated and control observations in the event period relative to that difference in the reference period. Model (1) was used to examine potential violations of the parallel trends assumption [37,38]. In addition, our study design assumes no unmeasured time-varying confounding and no residual confounding, after weighting, by fixed differences between upazilas within divisions [36,52].

We also estimated the overall DD effect using a standard 2-way fixed effects model that replaced the event periods in Eq 1 with a time-varying indicator,  $A_{ct-0.75}$ , equal to 1 if the observation was from a treated upazila after it had gained access to the MHVS, and lagged by 9 months since observations measured directly after the upazila gained access to the MHVS would not have been eligible for the program.

To assess the robustness of our main findings, we compared our main event study estimates with a 2.5 km buffer around each treated upazila to those with no buffer and a larger 5 km buffer. Additionally, we compared our main results from weighted event study models to unweighted estimates. All models incorporated cluster robust standard errors to account for clustering of observations within upazilas [53].

#### Results

Among women 15 to 49 years of age at the time of their interview, we restricted our samples to pregnancies and live births occurring in the 5 years prior to interview during the study period from 2000 to 2016; sample sizes ranged from 24,699 live births for analyses of antenatal care to 37,909 pregnancies lasting at least 7 months for analyses of stillbirth (S1 Table), from women in 508 upazilas (52 treated and 456 control). After excluding observations with missing outcome or covariate information (ranging from 2.8% to 3.3% across outcomes), sample sizes for complete case analyses ranged from 23,947 live births for analyses of antenatal care to 36,677 pregnancies for analyses of stillbirth.

Sample characteristics for live births and pregnancies prior to December 2006, when the MHVS was first introduced in treated upazilas, are presented in **Tables 2** and **3**, respectively. In general, women from areas that gained access to the MHVS were more socioeconomically disadvantaged, in terms of educational attainment and household wealth, and were more likely to come from rural areas. Women from treated areas were also less likely to report use of health services than women in control areas, including at least 3 antenatal care visits (22.2% versus 32.8%), delivering in an institution or health facility (7.2% versus 15.3%), and having a skilled birth attendant at delivery (10.8% versus 19.2%). Treated and control groups were similar in their distributions of neonatal mortality (4.0% in both groups), infant mortality (5.8% among treated versus 6.4% among control), and stillbirth (3.5% among treated versus 3.2% among control). Trends in maternal health services utilization, stillbirth, and neonatal and infant mortality before, during, and after MHVS implementation are shown in Fig 2.

After excluding upazilas outside the region of common support (**Appendix B** in **S1 Appendix**), weighted analyses included 338 upazilas (50 treated, 288 control) that were balanced in terms of measured upazila-level sociodemographic characteristics (the average standardized mean difference was reduced from 46.2% in the unweighted sample to 3.2% in the weighted

	Total			Control			Treated		
	N	Mean	SD	n	Mean	SD	n	Mean	SD
Household size	11,315	6.5	3.2	9,698	6.5	3.2	1617	6.5	3.1
Maternal age at marriage	11,315	15.8	2.8	9,698	15.9	2.8	1617	15.7	2.6
Respondent's age at event <sup>1</sup>	11,315	23.8	6.1	9,698	23.8	6.1	1617	23.9	6.2
	N	Frequency	%	N	Frequency	%	N	Frequency	%
Rural residence	11,315	7,746	68.5	9,698	6,410	66.1	1,617	1,336	82.6
Mother's education									
No schooling	11,315	3,649	32.2	9,698	3,061	31.6	1,617	588	36.4
Primary	11,315	3,577	31.6	9,698	3,034	31.3	1,617	543	33.6
Secondary or more	11,315	4,087	36.1	9,698	3,601	37.1	1,617	486	30.1
Father's education									
No schooling	11,315	4,131	36.5	9,698	3,470	35.8	1,617	661	40.9
Primary	11,315	3,159	27.9	9,698	2,665	27.5	1,617	494	30.6
Secondary or more	11,315	4,016	35.5	9,698	3,555	36.7	1,617	461	28.5
Household wealth >median	11,315	5,351	47.3	9,698	4,797	49.5	1,617	554	34.3
Birth interval <24 months <sup>2</sup>	11,315	1,278	11.3	9,698	1,082	11.2	1,617	196	12.1
3+ antenatal care visits	8,978	2,809	31.3	7,714	2,528	32.8	1,264	281	22.2
Institutional delivery	11,283	1,596	14.1	9,671	1,480	15.3	1,612	116	7.2
Skilled birth attendant	11,306	2,032	18.0	9,690	1,857	19.2	1,616	175	10.8
Neonatal mortality	11,248	448	4.0	9,639	384	4.0	1,609	64	4.0
Infant mortality	9,481	595	6.3	8,125	517	6.4	1,356	78	5.8

Table 2. Sample characteristics for analyses of live births from the period prior to the introduction of the MHVS in treated upazilas (before December 2006); BDHS data.

<sup>1</sup>Women's age at the time of the live birth, measured continuously.

<sup>2</sup>Binary indicator for whether the interval between the index birth and a prior birth was short (<24 months) or not (24+ months and first births). BDHS, Bangladesh Demographic and Health Survey; MHVS, Maternal Health Voucher Scheme; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004022.t002

sample, as shown in <u>S3 Table</u>). Weighted DD estimates from event study models are shown in <u>Fig 3</u>, with corresponding estimates in <u>S4 Table</u>. Each point represents the difference in the probability of the outcome comparing treated and control observations in the event period (e.g., 2 to 4 years after implementation) relative to the reference period immediately prior to implementation, along with robust 95% confidence intervals (CI).

Preintervention trends were approximately parallel for all outcomes except delivery by cesarean section, which are shown in **Appendix C** in **S1 Appendix**. In the postintervention period, there was evidence of increases in the use of maternal health services that materialized several years after the introduction of the MHVS, suggesting a lagged association. For example, after 6 years of access to the MHVS, the probabilities of delivering in a health facility and having a skilled birth attendant present increased by 6.5 (95% CI = -0.6, 13.6) and 5.8 (95% CI = -1.8, 13.3) percentage points, respectively, in treated areas compared to the control group. There was less evidence for a sustained increase in the receipt of antenatal care; 6 years after program implementation, the probability of reporting at least 3 antenatal care visits increased by 3.0 (95% CI = -4.8, 10.7) percentage points. We did not observe evidence consistent with the program having an impact on the probabilities of stillbirth, neonatal mortality, or infant mortality, with probabilities of stillbirth, neonatal mortality, and infant mortality decreasing by 0.7 (95% CI = -1.3, 2.6), 0.8 (95% CI = -1.7, 3.4), and 1.3 (95% CI = -2.5, 5.1) percentage points, respectively, after 6 years of access to the MHVS. Unweighted estimates were

	Total			Control			Treated		
	N	Mean	SD	n	Mean	SD	n	Mean	SD
Household size	11,977	6.5	3.2	10,264	6.5	3.2	1,713	6.4	3.1
Maternal age at marriage	11,977	15.8	2.8	10,264	15.9	2.8	1,713	15.7	2.6
Respondent's age at event <sup>1</sup>	11,977	23.7	6.1	10,264	23.7	6.1	1,713	23.8	6.1
	N	Frequency	%	N	Frequency	%	N	Frequency	%
Rural residence	11,977	8,192	68.4	10,264	6,774	66.0	1,713	1,418	82.8
Mother's education									
No schooling	11,977	3,815	31.9	10,264	3,190	31.1	1,713	625	36.5
Primary	11,977	3,778	31.5	10,264	3,209	31.3	1,713	569	33.2
Secondary or more	11,977	4,382	36.6	10,264	3,863	37.6	1,713	519	30.3
Father's education									
No schooling	11,977	4,363	36.4	10,264	3,658	35.6	1,713	705	41.2
Primary	11,977	3,330	27.8	10,264	2,812	27.4	1,713	518	30.2
Secondary or more	11,977	4,273	35.7	10,264	3,784	36.9	1,713	489	28.5
Household wealth >median	11,977	5,916	49.4	10,264	5,283	51.5	1,713	633	37.0
Birth interval <24 months <sup>2</sup>	11,977	1,334	11.1	10,264	1,133	11.0	1,713	201	11.7
Prior stillbirth	11,977	250	2.1	10,264	220	2.1	1,713	30	1.8
Stillbirth	11,977	386	3.2	10,264	326	3.2	1,713	60	3.5

Table 3. Sample characteristics for analyses of pregnancies lasting at least 7 months from the period prior to the introduction of the MHVS in treated upazilas (before December 2006); BDHS data.

<sup>1</sup>Women's age at the time of the stillbirth or live birth, measured continuously.

<sup>2</sup>Binary indicator for whether the interval between the index pregnancy outcome and a prior birth was short (<24 months) or not (24+ months and first births). BDHS, Bangladesh Demographic and Health Survey; MHVS, Maternal Health Voucher Scheme; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004022.t003

qualitatively similar (S1 Fig), as were weighted estimates with different buffers (S2 and S3 Figs) and standard DD estimates (Fig 4).

#### Discussion

Bangladesh's MHVS, introduced in 2007 and subsequently expanded to roughly 10% of the nation's subdistricts, is one of many health care voucher programs that have proliferated across low- and middle-income countries (LMICs) over the past 15 years [30]. Our analyses support 2 main conclusions. First, the introduction of the MHVS was positively associated with selected maternal health services, particularly the probability of delivering in a health facility, which is consistent with the literature on the impact of voucher schemes in LMICs [33], including Bangladesh [20,23,24,27,28]. Second, despite having a longer period of follow-up than most extant evaluations [31,33], we did not observe attendant reductions in stillbirth, neonatal mortality, or infant mortality.

Our results cohere with the few empirical studies that have examined impacts of voucher programs on measures of utilization and health outcomes [33], with prior research indicating that access to the MHVS was associated with greater use of maternal health services, including antenatal care, having a skilled attendant at delivery, delivering in a health facility, receiving postnatal care, and more comprehensive maternal care [20,24,25,27]. One cross-sectional study comparing voucher recipients to nonrecipients suggested that recipients were more likely to seek medical assistance in case of obstetric complications [54]. However, as with the broader literature on maternal voucher programs, the evidence for impacts of voucher programs on perinatal, infant, and maternal health outcomes, including mortality, is limited [30,33].



Fig 2. Trends in measures of maternal health services utilization, stillbirth, and neonatal and infant mortality among Bangladesh MHVS respondents, 2000–2016. MHVS, Maternal Health Voucher Scheme.

https://doi.org/10.1371/journal.pmed.1004022.g002

One potential explanation for why robust increases in institutional delivery and modest increases in antenatal care and skilled delivery have not been clearly associated with improved health outcomes relates to the targeting and implementation of the MHVS. Although awareness of the MHVS was high in some studies [20], it was lower in others [21,29], which may have reduced demand and utilization of the program. National reports suggest that the program distributed vouchers to 72% of the estimated number of potential beneficiaries between 2011 and 2016 [21]. However, although eligibility criteria related to pregnancy and household income were not strictly enforced [20,21,23], by design, the program may have excluded women at greater risk of adverse pregnancy and birth outcomes, including those lacking access to family planning [55].

Second, increased utilization of maternal health services may be insufficient to bring about improvements in health indicators if the quality of care remains low [56]. At least initially, the program failed to induce nonpublic providers to enroll in the program [20,23]; this may have limited competition between providers, which is a key mechanism through which demandside funding schemes are theorized to reduce costs and increase quality. Additionally, reports indicate that providers have not been adequately compensated for services rendered or increased workload, which may have reduced the quality of available services [12,20,21,23]. Among enrolled clinics, shortages of facilities, services (i.e., emergency obstetric care), staff, and specialists were recurring issues [21,23]. These challenges, while not unique to the MHVS in Bangladesh [57], may have limited the potential for the program to have its intended effects.

Third, there may be behavioral and sociocultural explanations that warrant further investigation. For example, recent work suggests that women who delivered in health facilities were



Fig 3. Weighted event study estimates of the effect of gaining access to the MHVS in specified periods before and after the implementation of the program; BDHS data, 2000-2016. Notes: Models included fixed effects for 6 divisions, fixed effects for year of birth (or the pregnancy outcome in analyses of stillbirth), and a vector of timevarying individual-level covariates, including household size, maternal age at marriage, women's age at the time of the stillbirth or live birth, rural residence, women's and husband's educational attainment, household wealth (above or below median), whether the interval between the index pregnancy or birth outcome and a prior birth was short (<24 months) or not (24+ months and first births), and prior stillbirth (in analyses of stillbirth). Weights accounted for the following upazila-level characteristics described in S2 Table: age structure, rural population, literacy, educational attainment, school attendance, poverty, employment, and household characteristics. Event periods included 4 leads (i.e., 6+, 4-6, 2-4, and 0-2 years prior to the introduction of the MHVS) and 4 lags (i.e., 0-2, 2-4, 4-6, and 6+ years after the introduction of the MHVS), with the 2-year period prior to implementation taken as the reference. Each coefficient represents the difference in the probability of the outcome (percentage point differences) comparing treated and control observations in the event period relative to the reference period; 95% confidence intervals account for the clustering of observations within upazilas. Sample sizes are reported in S1 Table, and the estimation of weights is described in Appendix B in S1 Appendix. BDHS, Bangladesh Demographic and Health Survey; MHVS, Maternal Health Voucher Scheme.

https://doi.org/10.1371/journal.pmed.1004022.g003

less likely than those who delivered in their homes to initiate breastfeeding within 1 hour of delivery [58], which may counteract positive effects of increased maternal health services use on infant survival through other mechanisms.

Finally, given our sample size, we were unable to detect small associations with adequate precision. An observational study comparing deliveries in 8 health facilities from treated upazilas with 8 facilities in matched control upazilas found that the incidence of stillbirth was 0.9 percentage points lower in treated facilities and concluded that "the incidence of stillbirths was significantly lower in voucher than in control areas" [20]. We found a similar effect size 6 years postimplementation, although with less precision, since our empirical approach trades off some precision in order to limit confounding [59]. Secondary data analyses that leverage existing demographic data can be viewed as a cost-effective way of generating evidence that can be triangulated with existing research.

There were other limitations to our study. With respect to measurement, the random geographic displacement of EAs may have led to nondifferential misclassification of the treatment, as described in **Appendix A** in **S1 Appendix**; in our main analyses, we added a buffer around treated upazilas that would reduce the misclassification of treated upazilas as control units and mitigate potential spillover effects, and tested different buffers from 0 to 5 km in sensitivity



**Fig 4. DD estimates comparing changes in the treated group before and after receiving access to the MHVS relative to corresponding trends in the control group; BDHS data, 2000–2016.** *Notes*: Models included fixed effects for 6 divisions, fixed effects for year of birth (or the pregnancy outcome in analyses of stillbirth), and a vector of time-varying individual-level covariates, including household size, maternal age at marriage, women's age at the time of the stillbirth or live birth, rural residence, women's and husband's educational attainment, household wealth (above or below median), whether the interval between the index pregnancy or birth outcome and a prior birth was short (<24 months) or not (24+ months and first births), and prior stillbirth (in analyses of stillbirth). Weights accounted for the following upazila-level characteristics described in S2 Table: age structure, rural population, literacy, educational attainment, school attendance, poverty, employment, and household characteristics; 95% confidence intervals account for the clustering of observations within upazilas. Sample sizes are reported in S1 Table, and the estimation of weights is described in S1 Appendix. BDHS, Bangladesh Demographic and Health Survey; DD, difference-in-difference; MHVS, Maternal Health Voucher Scheme.

https://doi.org/10.1371/journal.pmed.1004022.g004

analyses. Additionally, the locations of respondents' EAs were measured at the time of interview, which does not account for migration between the time of each reported birth/pregnancy outcome and the time of interview. Measurement of covariates and outcomes was based on self-report, and there is likely error that is nondifferential by treatment group. For outcomes that are likely underreported, such as stillbirth, this error would tend to bias our estimates only if it were differential between our treatment and control groups. With respect to confounding bias, there is the potential for unmeasured confounding, as in any nonrandomized study. However, we included fixed effects for treatment group, year, and division and accounted for measured upazila-level characteristics using inverse probability weights to create more exchangeable treatment groups. Nonetheless, incorporating weights with event study models likely reduced the precision of our estimates and we cannot rule out the possibility of reverse confounding (e.g., if access was targeted to areas with lower levels of access to maternal health care services) or residual confounding by public, private, or nongovernmental services or initiatives that coincided with the rollout of the MHVS, differentially in treated and control upazilas. If, for example, the existing supply of community health workers in each area influenced the process by which the MHVS was phased-in across upazilas and magnified its impact on health care services use, then the effect of providing access to the program may have been overestimated. Finally, regarding the external validity of our estimates, we selected our analytical sample based on the eligibility criteria of the MHVS, with further restriction in our main analyses to the area of common support based on propensity score distributions. We did not apply

DHS survey sampling weights and any inference beyond our sample should be done cautiously [60].

The aim of this quasi-experimental study was to provide a more comprehensive, longerterm, population-level analysis of the MHVS. Caveats considered, our results suggest that the MHVS was associated with increased utilization of maternal health services, particularly delivery in a health facility, which was incentivized by the program. However, we did not detect evidence of improvements in birth, neonatal, and infant health outcomes. Potential explanations include the suboptimal implementation of the program and inadequate supply-side investments to match increased demand and ensure access to quality health services. These challenges should be investigated and addressed [21] if the MHVS is to accelerate Bangladesh's progress toward achieving its maternal and newborn health goals.

#### Supporting information

S1 Checklist. RECORD statement: Checklist of items, extended from the STROBE statement, that should be reported in observational studies using routinely collected health data.

(DOCX)

**S1 Protocol. Prospective analysis plan.** (PDF)

**S1** Appendix. Appendix A, Appendix B, and Appendix C. (DOCX)

**S1 Table. Sample sizes for primary study outcomes.** (DOCX)

S2 Table. Upazila-level characteristics from the World Bank's Bangladesh Interactive Poverty Maps.

(DOCX)

S3 Table. Standardized mean differences comparing upazila-level sociodemographic factors in the unweighted and stabilized inverse probability of treatment weighted samples. (DOCX)

**S4** Table. Difference-in-differences estimates from events study models. (DOCX)

S1 Fig. Unweighted event study estimates of the effect of gaining access to the Maternal Health Voucher Scheme in specified periods before and after the implementation of the program; Bangladesh Demographic and Health Survey Data, 2000–2016. (DOCX)

S2 Fig. Weighted event study estimates of the effect of gaining access to the Maternal Health Voucher Scheme in specified periods before and after the implementation of the program; sensitivity analyses with 0-kilometer buffer around treated upazilas; Bangladesh Demographic and Health Survey Data, 2000–2016. (DOCX)

S3 Fig. Weighted event study estimates of the effect of gaining access to the Maternal Health Voucher Scheme in specified periods before and after the implementation of the program; sensitivity analyses with 5-kilometer buffer around treated upazilas; Bangladesh **Demographic and Health Survey Data, 2000–2016.** (DOCX)

#### **Author Contributions**

**Conceptualization:** Arijit Nandi, Thomas J. Charters, Amm Quamruzzaman, Erin C. Strumpf, Jay S. Kaufman, Jody Heymann, Arnab Mukherji, Sam Harper.

Data curation: Arijit Nandi, Thomas J. Charters, Amm Quamruzzaman.

Formal analysis: Arijit Nandi, Thomas J. Charters, Sam Harper.

Funding acquisition: Arijit Nandi.

**Investigation:** Arijit Nandi, Thomas J. Charters, Amm Quamruzzaman, Erin C. Strumpf, Jay S. Kaufman, Jody Heymann, Arnab Mukherji, Sam Harper.

Methodology: Arijit Nandi, Thomas J. Charters, Amm Quamruzzaman, Erin C. Strumpf, Jay S. Kaufman, Jody Heymann, Arnab Mukherji, Sam Harper.

Project administration: Arijit Nandi.

Resources: Arijit Nandi.

Supervision: Arijit Nandi.

Validation: Arijit Nandi.

Writing - original draft: Arijit Nandi, Thomas J. Charters.

Writing – review & editing: Arijit Nandi, Thomas J. Charters, Amm Quamruzzaman, Erin C. Strumpf, Jay S. Kaufman, Jody Heymann, Arnab Mukherji, Sam Harper.

#### References

- Chowdhury AMR, Bhuiya A, Chowdhury ME, Rasheed S, Hussain Z, Chen LC. The Bangladesh paradox: exceptional health achievement despite economic poverty. Lancet. 2013; 382(9906):1734–1745. https://doi.org/10.1016/S0140-6736(13)62148-0 PMID: 24268002
- Adams AM, Ahmed T, El Arifeen S, Evans TG, Huda T, Reichenbach L. Innovation for universal health coverage in Bangladesh: a call to action. Lancet. 2013; 382(9910):2104–2111. https://doi.org/10.1016/ S0140-6736(13)62150-9 PMID: 24268605
- El Arifeen S, Hill K, Ahsan KZ, Jamil K, Nahar Q, Streatfield PK. Maternal mortality in Bangladesh: a Countdown to 2015 country case study. Lancet. 2014; 384(9951):1366–1374. https://doi.org/10.1016/ S0140-6736(14)60955-7 PMID: 24990814
- United Nations Inter-agency Group for Child Mortality Estimation (UN IGME), 'Levels & Trends in Child Mortality: Report 2019, Estimates developed by the United Nations Inter-agency Group for Child Mortality Estimation'. New York: United Nations Children's Fund; 2019.
- 5. Trends in maternal mortality 2000 to 2017: estimates by WHO, UNICEF, UNFPA, World Bank Group and the United Nations Population Division Geneva: World Health Organization; 2019.
- Hajizadeh M, Alam N, Nandi A. Social inequalities in the utilization of maternal care in Bangladesh: Have they widened or narrowed in recent years? Int J Equity Health. 2014; 13(1):120. https://doi.org/10. 1186/s12939-014-0120-4 PMID: 25492512
- Kamal N, Curtis S, Hasan MS, Jamil K. Trends in equity in use of maternal health services in urban and rural Bangladesh. Int J Equity Health. 2016; 15:27–. <u>https://doi.org/10.1186/s12939-016-0311-2</u> PMID: 26883742.
- Rahman MH, Mosley WH, Ahmed S, Akhter HH. Does service accessibility reduce socioeconomic differentials in maternity care seeking? Evidence from rural Bangladesh. J Biosoc Sci. 2008; 40(1):19–33. https://doi.org/10.1017/S0021932007002258 PMID: 17588280
- National Institute of Population Research and Training (NIPORT), and ICF. Bangladesh Demographic and Health Survey 2017–18. Dhaka, Bangladesh, and Rockville, Maryland, USA: NIPORT and ICF; 2020.

- Schmidt J-O, Ensor T, Hossain A, Khan S. Vouchers as demand side financing instruments for health care: a review of the Bangladesh maternal voucher scheme. Health Policy. 2010; 96(2):98–107. <a href="https://doi.org/10.1016/j.healthpol.2010.01.008">https://doi.org/10.1016/j.healthpol.2010.01.008</a> PMID: 20138385
- Anwar I, Sami M, Akhtar N, Chowdhury M, Salma U, Rahman M, et al. Inequity in maternal health-care services: evidence from home-based skilled-birth-attendant programmes in Bangladesh. Bull World Health Organ. 2008; 86:252–259. https://doi.org/10.2471/blt.07.042754 PMID: 18438513
- Ahmed S, Khan MM. A maternal health voucher scheme: what have we learned from the demand-side financing scheme in Bangladesh? Health Policy Plan. 2011; 26(1):25–32. https://doi.org/10.1093/ heapol/czq015 PMID: 20375105
- Campbell J, Dussault G, Buchan J, Pozo-Martin F, Guerra Arias M, Leone C, et al. A universal truth: no health without a workforce. Forum Report, Third Global Forum on Human Resources for Health, Recife, Brazil. Geneva: Global Health Workforce Alliance and World Health Organization; 2013.
- Akter S, Davies K, Rich JL, Inder KJ. Barriers to accessing maternal health care services in the Chittagong Hill Tracts, Bangladesh: A qualitative descriptive study of Indigenous women's experiences. PIoS ONE. 2020; 15(8):e0237002. https://doi.org/10.1371/journal.pone.0237002 PMID: 32780774
- Choudhury N, Ahmed SM. Maternal care practices among the ultra poor households in rural Bangladesh: a qualitative exploratory study. BMC Pregnancy Childbirth. 2011; 11(1):1–8. https://doi.org/10. 1186/1471-2393-11-15 PMID: 21362164
- Rahman MM, Rob U, Noor FR, Bellows B. Out-of-pocket expenses for maternity care in rural Bangladesh: a public-private comparison. Int Q Community Health Educ. 2013; 33(2):143–157.
- Keya KT, Rob U, Rahman MM, Bajracharya A, Bellows B. Distance, transportation cost, and mode of transport in the utilization of facility-based maternity services: evidence from rural Bangladesh. Int Q Community Health Educ. 2014; 35(1):37–51. https://doi.org/10.2190/IQ.35.1.d PMID: 25416431
- 18. Rannan-Eliya R, Kasthuri G, Begum T, Rahman A, Hossain N, Anuranga C. Impact of maternal and child health private expenditure on poverty and inequity in Bangladesh: out-of-pocket payments by patients at Ministry of Health and Family Welfare Facilities in Bangladesh and the impact of the maternal voucher scheme on costs and access of mothers and children; technical report B. Mandaluyong City, Philippines: Asian Development Bank; 2012.
- Nahar S, Costello A. The hidden cost of 'free'maternity care in Dhaka, Bangladesh. Health Policy Plan. 1998; 13(4):417–422. https://doi.org/10.1093/heapol/13.4.417 PMID: 10346033
- 20. Hatt L, Nguyen H, Sloan N, Miner S, Magvanjav O, Sharma A, et al. Economic evaluation of demandside financing (DSF) for maternal health in Bangladesh. Bethesda, Maryland: Review, Analysis and Assessment of Issues Related to Health Care Financing and Health Economics in Bangladesh, Abt Associates Inc; 2010.
- 21. Khan MM, Khan AR. Report on the Diagnostic Study of Demand Side Financing—Maternal Health Voucher Scheme of Bangladesh. Dhaka, Bangladesh: Maxwell Stamp PLC; 2016.
- Hasan S. Upazila development planning in Bangladesh: Problems of resource mobilization. Asian Survey. 1992; 32(9):802–814.
- Koehlmoos TLP, Ashraf A, Kabir H, Islam Z, Gazi R, Saha NC, et al. Rapid assessment of demand-side financing experiences in Bangladesh. Bangladesh: International Centre for Diarrhoeal Disease Research, Bangladesh; 2008.
- Mahmood SS, Amos M, Hoque S, Mia MN, Chowdhury AH, Hanifi SMA, et al. Does healthcare voucher provision improve utilisation in the continuum of maternal care for poor pregnant women? Experience from Bangladesh. Global Health Action. 2019; 12(1):1701324. <u>https://doi.org/10.1080/16549716.2019</u>. 1701324 PMID: 31825301
- 25. Mia MN, Mahmood SS, Iqbal M, Bhuiya A, Pallikadavath S, Stones RW. The Bangladesh Maternal Health Voucher Scheme: impact on completeness of antenatal care provision. J Biosoc Sci. 2020.
- **26.** Das AC. Improving access to safe delivery for poor women by voucher scheme in Bangladesh. South East Asia J Public Health. 2015; 5(1):39–43.
- Nguyen HT, Hatt L, Islam M, Sloan NL, Chowdhury J, Schmidt J-O, et al. Encouraging maternal health service utilization: an evaluation of the Bangladesh voucher program. Soc Sci Med. 2012; 74(7):989– 996. https://doi.org/10.1016/j.socscimed.2011.11.030 PMID: 22326107
- Keya KT, Bellows B, Rob U, Warren C. Improving Access to Delivery Care and Reducing the Equity Gap Through Voucher Program in Bangladesh: Evidence From Difference-in-Differences Analysis. Int Q Community Health Educ. 2018; 38(2):137–145. <u>https://doi.org/10.1177/0272684X17749568</u> PMID: 29298635
- 29. Talukder M, Rob U, Musa SAJM, Bajracharya A, Keya KT, Noor FR, et al. Evaluation of the impact of the voucher program for improving maternal health behavior and status in Bangladesh. Dhaka: Population Council; 2014.

- Bellows BW, Conlon CM, Higgs ES, Townsend JW, Nahed MG, Cavanaugh K, et al. A taxonomy and results from a comprehensive review of 28 maternal health voucher programmes. J Health Popul Nutr. 2013; 31(4 SUPPL.2):S106–S128. PMID: 24992806.
- Hunter BM, Harrison S, Portela A, Bick D. The effects of cash transfers and vouchers on the use and quality of maternity care services: a systematic review. PLoS ONE. 2017; 12(3):e0173068. <a href="https://doi.org/10.1371/journal.pone.0173068">https://doi.org/10.1371/journal.pone.0173068</a> PMID: 28328940
- Murray SF, Hunter BM, Bisht R, Ensor T, Bick D. Effects of demand-side financing on utilisation, experiences and outcomes of maternity care in low-and middle-income countries: a systematic review. BMC Pregnancy Childbirth. 2014; 14(1):30. https://doi.org/10.1186/1471-2393-14-30 PMID: 24438560
- Hurst TE, Semrau K, Gawande A, Hirschhorn LR. Demand-side interventions for maternal care: evidence of more use, not better outcomes. BMC Pregnancy Childbirth. 2015; 15(1):297.
- 34. Angrist JD, Pischke J-S. Mostly harmless econometrics: An empiricist's companion: Princeton University Press; 2008.
- **35.** Shadish WR, Cook TD, Campbell DT. Experimental and quasi-experimental designs for generalized causal inference: Wadsworth Cengage Learning; 2002.
- Strumpf EC, Harper S, Kaufman JS. Fixed effects and difference in differences. Methods in Social Epidemiology: Jossey-Bass, San Francisco, CA; 2017.
- Goodman-Bacon A. Difference-in-differences with variation in treatment timing. National Bureau of Economic Research. 2018:0898–2937.
- Jacobson LS, LaLonde RJ, Sullivan DG. Earnings losses of displaced workers. Am Econ Rev. 1993;685–709.
- Corsi DJ, Neuman M, Finlay JE, Subramanian SV. Demographic and health surveys: a profile. Int J Epidemiol. 2012; 41(6):1602–13. Epub 2012/11/14. https://doi.org/10.1093/ije/dys184 PMID: 23148108.
- 40. National Institute of Population Research and Training (NIPORT), Mitra and Associates, and ICF International. Bangladesh Demographic and Health Survey 2014. Dhaka, Bangladesh, and Rockville, Maryland, USA: NIPORT, Mitra and Associates, and ICF International; 2016.
- 41. National Institute of Population Research and Training (NIPORT), Mitra and Associates, and ICF International. Bangladesh Demographic and Health Survey 2011. Dhaka, Bangladesh, and Calverton, Maryland, USA: NIPORT, Mitra and Associates, and ICF International; 2013.
- 42. National Institute of Population Research and Training (NIPORT), Mitra and Associates, and Macro International. Bangladesh Demographic and Health Survey 2007. Dhaka, Bangladesh, and Calverton, Maryland, USA: NIPORT, Mitra and Associates, and Macro International; 2009.
- **43.** National Institute of Population Research and Training (NIPORT), Mitra and Associates, and ORC Macro. Bangladesh Demographic and Health Survey 2004. Dhaka, Bangladesh, and Calverton, Maryland, USA: NIPORT, Mitra and Associates, and ORC Macro; 2005.
- 44. World Bank. Bangladesh Interactive Poverty Maps. November 10, 2016 [cited 2020 May 15]. Available from: https://www.worldbank.org/en/data/interactive/2016/11/10/bangladesh-poverty-maps.
- Islam N, Islam MT, Yoshimura Y. Practices and determinants of delivery by skilled birth attendants in Bangladesh. Reprod Health. 2014; 11(1):86. https://doi.org/10.1186/1742-4755-11-86 PMID: 25495451
- Al Kibria GM, Ghosh S, Hossen S, Barsha RAA, Sharmeen A, Uddin SI. Factors affecting deliveries attended by skilled birth attendants in Bangladesh. Matern Health Neonatol Perinatol. 2017; 3(1):7.
- **47.** World Health Organization. Definitions and indicators in family planning maternal & child health and reproductive health used in the WHO regional office for Europe. Copenhagen: WHO Regional Office for Europe; 2000.
- Filmer D, Pritchett L. The Effect of Household Wealth on Educational Attainment: Evidence from 35 Countries. Popul Dev Rev. 1999; 25(1):85–120. https://doi.org/10.1111/j.1728-4457.1999.00085.x
- 49. Filmer D, Pritchett LH. Estimating wealth effects without expenditure data—or tears: an application to educational enrollments in states of India. Demography. 2001; 38(1):115–132. <u>https://doi.org/10.1353/dem.2001.0003</u> PMID: 11227840
- Austin PC, Stuart EA. Moving towards best practice when using inverse probability of treatment weighting (IPTW) using the propensity score to estimate causal treatment effects in observational studies. Stat Med. 2015; 34(28):3661–3679. https://doi.org/10.1002/sim.6607 PMID: 26238958
- Garrido MM, Kelley AS, Paris J, Roza K, Meier DE, Morrison RS, et al. Methods for constructing and assessing propensity scores. Health Serv Res. 2014; 49(5):1701–1720. <u>https://doi.org/10.1111/1475-6773.12182</u> PMID: 24779867
- 52. Meyer BD. Natural and quasi-experiments in economics. J Bus Econ Stat. 1995; 13(2):151–161.

- Cameron AC, Miller DL. A practitioner's guide to cluster-robust inference. J Hum Resour. 2015; 50 (2):317–372.
- Ahmed S, Khan MM. Is demand-side financing equity enhancing? Lessons from a maternal health voucher scheme in Bangladesh. Soc Sci Med. 2011; 72(10):1704–1710. https://doi.org/10.1016/j. socscimed.2011.03.031 PMID: 21546145.
- 55. Tsui AO, McDonald-Mosley R, Burke AE. Family planning and the burden of unintended pregnancies. Epidemiol Rev. 2010; 32(1):152–174. https://doi.org/10.1093/epirev/mxq012 PMID: 20570955
- Kruk ME, Larson E, Twum-Danso NA. Time for a quality revolution in global health. Lancet Glob Health. 2016; 4(9):e594–e596. https://doi.org/10.1016/S2214-109X(16)30131-0 PMID: 27539798
- 57. Jehan K, Sidney K, Smith H, De Costa A. Improving access to maternity services: an overview of cash transfer and voucher schemes in South Asia. Reprod Health Matters. 2012; 20(39):142–154. <u>https://doi.org/10.1016/S0968-8080(12)39609-2</u> PMID: 22789092
- Karim F, Khan ANS, Tasnim F, Chowdhury MAK, Billah SM, Karim T, et al. Prevalence and determinants of initiation of breastfeeding within one hour of birth: An analysis of the Bangladesh Demographic and Health Survey, 2014. PloS ONE. 2019; 14(7):e0220224. <u>https://doi.org/10.1371/journal.pone.</u> 0220224 PMID: 31344069
- Kaufman JS. Commentary: Why are we biased against bias? Int J Epidemiol. 2008; 37(3):624–626. https://doi.org/10.1093/ije/dyn035 PMID: 18316349
- Platt RW, Harper SB. Survey data with sampling weights: is there a "best" approach. Environ Res. 2013; 120:143–144. https://doi.org/10.1016/j.envres.2012.08.006 PMID: 22995642